Author’s response to reviews

Title: Adherence to drug treatments and adjuvant barrier repair therapies are key factors for clinical improvement in mild to moderate acne: The ACTUO observational prospective multicenter cohort trial in 643 patients.

Authors:

Raul De Lucas (rauldelucas@gmail.com)
Gerardo Moreno-Arias (masmilan2000@yahoo.it)
Montserrat Perez-López (massimomilani1959@gmail.com)
Angel Vera-Casaño (massimomilani1959@hotmail.com)
Sonia Aladren (sonia.aladren@idin.com)
Massimo Milani (massimo.milani@isdin.com)

Version: 5 Date: 20 July 2015

Author’s response to reviews:

1. Authors Contributions - added

2. Title page - It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author. Added

3. Figure legends section - No title or legend should be attached to the figure image. Instead, could you please include a Figure Legends section to appear after the References section? ok DONE